Author's response to reviews

Title: Epigenetic alterations differ in phenotypically distinct human neuroblastoma cell lines

Authors:

Qiwei Yang (yangq@uchicago.edu)
Yufeng Tian (ytian@peds.bsd.uchicago.edu)
Kelly R Ostler (kostler@uchicago.edu)
Alexandre Chlenski (achlensk@peds.bsd.uchicago.edu)
Lisa J Guerrero (lguerre1@peds.bsd.uchicago.edu)
Helen R Salwen (hsalwen@peds.bsd.uchicago.edu)
Lucy A Godley (lgodley@medicine.bsd.uchicago.edu)
Susan L Cohn (scohn@peds.bsd.uchicago.edu)

Version: 5 Date: 11 June 2010

Author's response to reviews:

1. On page 9, Change “analyses was” to “analyses were”.
2. on page 10, change “This dose and time point have” to “This dose and time point has”.
3. on page 21, change nontumorigenic to non-tumorigenic.
4. modify figure 4 and figure 6